Author's response to reviews

Title: Neck muscle cross-sectional area, brain volume and cognition in healthy older men; a cohort study

Authors:

Alixe HM Kilgour (a.kilgour@ed.ac.uk)
Karen J Ferguson (karen.ferguson@ed.ac.uk)
Calum D Gray (calum.gray@ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)
Joanna M Wardlaw (joanna.wardlaw@ed.ac.uk)
Alasdair MJ MacLullich (a.maclullich@ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)

Version: 2 Date: 30 October 2012

Author's response to reviews:

Added "n" to the end of men in the title.
Thank you.